Author’s response to reviews

Title: Impact of stakeholder perspectives on cost-effectiveness estimates of four specialized nutritious foods for preventing stunting and wasting in children 6-23 months in Burkina Faso

Authors:

Ye Shen (ye.shen@tufts.edu; yeshen92@gmail.com)
Ilana Cliffer (Ilana.Cliffer@tufts.edu)
Devika Suri (Devika.Suri@tufts.edu)
Breanne Langlois (Breanne.Langlois@tufts.edu)
Stephen Vosti (vosti@primal.ucdavis.edu)
Patrick Webb (Patrick.Webb@tufts.edu)
Beatrice Rogers (beatrice.rogers@tufts.edu)

Version: 3 Date: 13 Feb 2020

Author’s response to reviews:

Dear Dr. Collins,

Thanks for flagging the error in the references. We have now corrected this error in the manuscript. Looking forward to moving this forward!

Best,

Ye